\def\papertitle{On the strongly mid p-summing operators and application}
\def\paperauthorA{Ferradi Athmane}
%\def\paperauthorB{Author_B_fullname}
% you can add more authors using alphabets after \paperauthor
% 
%------------------------------------------------------------------------------------------
% Do not delete this part starting from here and up to .....
\documentclass[oneside]{article}
\usepackage{icma2021,amssymb,amsmath,mathrsfs} 
%\setcounter{page}{1}
%\input{../../expages.tex}\setpagenumber{1}
%\input{expages.tex}\setpagenumber{1}
\ninept

\usepackage{mathptmx} 
% Saves a lot of ouptut space in PDF... after conversion with the distiller	  
% Delete if you cannot get PS fonts working on your system.

% pdf-tex settings: detect automatically if run by latex or pdflatex
\newif\ifpdf
\ifx\pdfoutput\relax
\else
\ifcase\pdfoutput
\pdffalse
\else
\pdftrue
\fi

\ifpdf % compiling with pdflatex
\usepackage[pdftex,
pdftitle={\papertitle},
%pdfauthor={\paperauthorA},
% for more than one author use pdfauthor={\paperauthorA,\paperauthorB, .....},
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%ù
colorlinks=false, % links are activated as colror boxes instead of color text
bookmarksnumbered, %use section numbers with bookmarks
pdfstartview= XYZ	% starts with zoom=100% instead of full screen; especially useful if working with a big screen :-)
]{hyperref}
\pdfcompresslevel=9
\usepackage[pdftex]{graphicx}
\usepackage[figure,table]{hypcap}
\else % compiling with latex
\usepackage[dvips]{graphicx}
\usepackage[dvips,
colorlinks=false,           % no color links
bookmarksnumbered, %use section numbers with bookmarks
pdfstartview= XYZ	% starts with zoom=100% instead of full screen
]{hyperref}
% hyperrefs are active in the pdf file after conversion
\usepackage[figure,table]{hypcap}
\fi

\newcommand\dsum{\displaystyle\sum}
\newcommand\dprod{\displaystyle\prod}
%------------------------------------------------------------------------------------------
\title{\papertitle}
\affiliation{\paperauthorA}	% This command replaces \name{The DAFx Crew} 
{University of M'sila}

%\usepackage{layout}
%------------------------------------------------------------------------------------------
%%%%%%%%%%%%%%%%%%%%%%% to here %%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
\begin{document}
\begin{thebibliography}{99}
\bibitem{transformation} G. Botelho and J.R. Campos. On the transformation of vector-valued sequences by multilinear operators, Monatsh. Math 183 (2017), 415-435.

\bibitem{mid1}G. Botelho, J.R. Campos and J. Santos, Operator ideals related to absolutely summing and Cohen strongly summing operators, Pacific J. Math. 287 (2017), 1–17.
\bibitem{DJT} J. Diestel, H. Jarchow and A. Tonge. Absolutely summing
operators. Cambridge University Press, Cambridge, (1995).

\bibitem{KS} A. Karn and D. Sinha, An operator summability of sequences in Banach spaces, Glasg. Math. J. 56 (2014), no. 2, 427–437.

\bibitem{Cohen} J. Cohen. Absolutely $p$-summing, $p$-nuclear operators and their conjugates, conjugates, Math. Ann 201 (1973), 177-200 .

\bibitem{Pie80} A. Pietsch. Operator ideals. Deutsch. Verlag Wiss, Berlin, 1978; North-Holland, Amsterdam-London-New York-Tokyo, (1980).

\bibitem{ab}Apiola, H.:  {Duality between spaces of $p$-summable sequences, $(p, q)$-summing operators and characterization of nuclearity.} Math. Ann \textbf{219}, 53–64 (1976).	
\end{thebibliography}

	
	%\layout
	
	% more pdf-tex settings:
	\ifpdf % used graphic file format for pdflatex
	\DeclareGraphicsExtensions{.png,.jpg,.pdf}
	\else  % used graphic file format for latex
	\DeclareGraphicsExtensions{.eps}
	\fi
	
	\maketitle
	
\begin{abstract}
In this paper, we introduce and study the new ideal of strongly mid $p$-summing linear operators between Banach spaces. We prove, an operator is strongly  mid $p$-summing if and only if its adjoint is absolutely mid $p^*$-summing. This result led us to prove, $X$ is $p$-Dunford-Pettis property $(1<p<\infty)$ if and only if, so is $X^*$.
\end{abstract}

\section{Notation and preliminaries}\label{section1}
The notation used in the paper is in general standard. The letters $X,X_1,\cdots,X_m, \ Y$ ($m$ be in $\mathbb{N}$ ) shall denote Banach spaces over  $\mathbb{K}$ (real or complex scalars field). We will denote by $ \mathcal{L}(X_{1},\cdots,X_{m};Y)$ the Banach space of all bounded multilinear operators from $X_{1}\times \cdots\times X_{m}$ into $Y$ equipped with the operator norm. $\mathcal{L}(X;Y) $ the Banach space of all bounded linear operators $T:X\rightarrow Y$ endowed with the usual sup norm. The closed unit ball of $X$  is denoted by $B_X$ and its topological dual by $X^*=\mathcal{L}(X;\mathbb{K})$.

For a Banach space $X$, and $1\leq p \leq \infty$, let $p^*$ be its conjugate that is $1/p+1/p^*=1$. Let us define the  sequences spaces we shall work with: (see \cite{DJT,mid1}).\\
$\ell_{p}(X)=$ the Banach space of absolutely $p$-summable sequences with the norm ($1\leq p < \infty$) $$\parallel\left(x_{j}\right)_{j=1}^{\infty}\parallel_{p}:=\left( \sum_{j=1}^{\infty}\parallel x_{j}\parallel^{p}\right) ^{1/p}.$$
$\ell_{p}^{w}(X)=$ the Banach space of weakly $p$-summable sequences with the norm ($1\leq p < \infty$) $$\parallel(x_{j})_{j=1}^{\infty}\parallel_{w, p}:= \sup _{x^* \in B_{X} *}\left(\sum_{j=1}^{\infty}\left|x^*\left(x_{j}\right)\right|^{p}\right) ^{1/p}.$$
$\ell_{p}^{mid}(X)$ the Banach space  of mid $p$-summable sequences with the norm ($1\leq p < \infty$) $$\parallel\left(x_{j}\right)_{j=1}^{\infty}\parallel_{mid, p}:=\sup _{\left(x_{n}^{*}\right)_{n=1}^{\infty} \in B_{\ell_p^w\left(X^{*}\right)}} \left( \sum_{j=1}^{\infty}\sum_{n=1}^{\infty}\left|x_{n}^{*}\left(x_{j}\right)\right|^p\right) ^{1/p}.$$
When $p=\infty$ we have $\parallel\left(x_{j}\right)_{j=1}^{\infty}\parallel_{\infty}=\parallel\left(x_{j}\right)_{j=1}^{\infty}\parallel_{w,\infty}=\parallel\left(x_{j}\right)_{j=1}^{\infty}\parallel_{mid,\infty}=\sup_j \Vert x_j\Vert$. The relationships between the various sequence spaces are given by \cite{mid1}
$$\ell_{p}(X)\subsetneq  \ell_{p}^{mid}(X)\subsetneq  \ell_{p}^{w}(X), $$with $$\parallel\left(x_{j}\right)_{j=1}^{\infty}\parallel_{p}\leq \parallel\left(x_{j}\right)_{j=1}^{\infty}\parallel_{w,p} \leq \parallel\left(x_{j}\right)_{j=1}^{\infty}\parallel_{mid,p}.$$

Recall that, for $1\leq p <\infty$, we say that a continuous linear operator $T:X\rightarrow Y$ is absolutely $p$-summing if $$(T(x_j))_{j=1}^\infty \in \ell_p(Y)   \   whenever \ (x_j)_{j=1}^\infty \in \ell^w_p(Y).$$ The class of absolutely $p$-summing operator from $X$ to $Y$ will be represented by $\Pi_p(X;Y)$. An equivalent formulation asserts that $T$ is absolutely $p$-summing if  there is a positive constant $C$ such that $$\Vert(T(x_j))_{j=1}^\infty \Vert_p \leq C \Vert(x_j)_{j=1}^\infty\Vert_{w,p}$$ for all $(x_j)_{j=1}^\infty\in \ell_p^w(X)$. The infimum of all $C>0$ that satisfy the above inequality define a norm, denoted by $ \pi_{p}(T)$ .

Let $(\mathcal{I}(X;Y),\alpha(\cdot))$ be an operator ideal, we put $$\mathcal{I}^d(X;Y)=\{ T\in \mathcal{L}(X;Y): T^*\in \mathcal{I}(Y^*;X^*) \}.$$ For an operator $T\in \mathcal{I}^d(X;Y)$, we put $\alpha^d(T)=\alpha(T^*)$. With these notations $(\mathcal{I}^d(X;Y),\alpha^d(\cdot))$ is also a Banach operator ideal and is called the dual ideal of $(\mathcal{I}(X;Y),\alpha(\cdot))$ (see \cite[Section 4]{Pie80}).

Let us recall now the definition of strongly $p$-summing linear operators. An operator $T\in \mathcal{L}(X;Y)$ is strongly $p$-summing if there is a constant $C$ such that  $$\Vert ( \langle T(x_j),y_j^*\rangle  ) _{j=1}^{\infty }\Vert _{1}\leq C\Vert
( x_{j}) _{j=1}^{\infty }\Vert _{p}\Vert (y_{j}^{\ast }) _{j=1}^{\infty }\Vert _{w,p^*}$$ for all $(x_j)_{j=1}^\infty\in \ell_p(X)$ and $(y^*_j)_{j=1}^\infty\in \ell_{p^*}^w(Y^*)$. The space $\mathcal{D}_p(X;Y)$ of all strongly $p$-summing linear operators from $X$ into $Y$ which is a Banach space with the norm $d_p(T)$  the infimum of all $C>0$ that satisfy the above inequality. According to \cite{Cohen,ab} we obtain $\mathcal{D}_p^d(X;Y)=\Pi_{p^*}(X;Y)$ and $\Pi_{p}^d(X;Y)=\mathcal{D}_{p^*}(X;Y)$.

Our results are presented as follows. In first section \ref{section1}, we recall important results and definitions to be used later.
In section \ref{section2}, we introduce and investigate a new ideal of strongly mid $p$-summing operators. We present a characterization given by a summability property. We also prove the related dual result: an operator $T\in \mathcal{L}(X;Y)$ is strongly mid $p$-summing ($T\in \mathcal{D}_p^{mid}(X;Y)$) if and only if its adjoint $T^*$ is absolutely mid $p^*$-summing ($T^*\in \Pi_{p^*}^{mid}(Y^*;X^*)$). We study Banach space $X$ for which $id_X\in \Pi_p^{mid}$(Theorem   \ref{theorema1}), $id_X\in \mathcal{D}_p^{mid}$(Theorem  \ref{theorema2}), $\mathcal{D}_{p}(\ell_p;X)=\mathcal{L}(\ell_p;X)$ (Theorem \ref{theorema4}). We also prove, $X$ has the $p$-Dunford-Pettis property if and only if $X^*$ has the $p$-Dunford-Pettis property.
\section{Strongly mid summing linear operators}\label{section2}
Botelho, Campos and Santos \cite{mid1} introduced the concept of absolutely mid $p$-summing operators. For $1\leq p<\infty$, an operator $T\in \mathcal{L}(X;Y)$ is absolutely mid $p$-summing if $$(T(x_j))_{j=1}^\infty\in \ell_p^{mid}(Y) \  whenever  \ (x_j)_{j=1}^\infty\in \ell_p(X).$$ By $\Pi_p^{mid}(X;Y)$, we denote the space of absolutely mid $p$-summing operators.

 We introduce the concept of strongly mid $p$-summing linear operators as a characterization of the conjugates of absolutely mid $p^*$-summing linear operators. This idea seems to have appeared for the first time in \cite{Cohen} and \cite{ab}.
\begin{definition}Let $1<p\leq \infty$. A mapping $T\in \mathcal{L}(X;Y)$ is strongly mid $p$-summing if there exist a constant $C>0$ such that 
\begin{equation}\label{defs2.1}
\Vert ( \langle T( x_{j}) ,y_{j}^*\rangle ) _{j=1}^{k}\Vert _{1}\leq C\Vert (x_{j}) _{j=1}^{k}\Vert _{p}\Vert ( y_{j}^*) _{j=1}^{k}\Vert _{mid,p^*}
\end{equation}
for any $( x_{j}) _{j=1}^{k}\subset X$ and $( y_{j}^{\ast })_{j=1}^{k}\subset Y^{\ast }.$ $\mathcal{D}_p^{mid}(X;Y)$ denotes the space of all  strongly mid $p$-summing operators from $X$ to $Y$. The least $C$  for which (\ref{defs2.1}) holds will be written $d_p^{mid}(T)$. From the definition it is clear that $\mathcal{D}_p(X;Y)\subset \mathcal{D}_p^{mid}(X;Y)$ and $d_p^{mid}(\cdot)\leq d_p(\cdot)$.
\end{definition}
For $T\in \mathcal{L}(X;Y)$. The induced map $\varphi_T:X\times Y^*\rightarrow \mathbb{K}$ given by $\varphi_T(x,y^*)=\langle T(x),y^*\rangle$ for all $x\in X$ and $y^*\in Y^*$ is continuous bilinear. Using  the abstract approach of \cite{transformation} and Proposition 1.9 in \cite{mid1}, we can see that the next proposition is immediate consequences of  \cite[Proposition 2.4]{transformation}.
\begin{proposition}Let $T\in \mathcal{L}(X;Y)$, the following assertions are equivalent.
\begin{enumerate}
\item $T\in \mathcal{D}_p^{mid}(X;Y)$.
\item $(\varphi_T (x_j,y^*_j))_{j=1}^{\infty } \in \ell_1$ whenever $(x_j)_{j=1}^{\infty }\in \ell_{ p}(X)$ and $(y^*_j)_{j=1}^{\infty }\in \ell_{ p^*}^{mid}(Y^*)$.
\item The induced map $\widehat{\varphi_T }:\ell _{p}(X) \times \ell	_{p^*}^{mid}(Y^*)    \rightarrow \ell_1 $ given by $\widehat{\varphi_T} ((x_{j}) _{j=1}^{\infty } ,(y^*_j)_{j=1}^{\infty })=(\varphi_T (x_j,y^*_j))_{j=1}^{\infty } $ is a well-define and continuous bilinear operator.
\item There exist a constant $C>0$ such that 
\begin{equation}\label{equ500}
\Vert ( \varphi_T(x_j,y_j^*) ) _{j=1}^{\infty }\Vert _{1}\leq C\Vert
( x_{j}) _{j=1}^{\infty }\Vert _{p}\Vert (y_{j}^{\ast }) _{j=1}^{\infty }\Vert _{mid,p^*}.
\end{equation}
for any $( x_{j}) _{j=1}^{\infty }\in \ell _{p}( X)$ and  $(y_{j}^{\ast }) _{j=1}^{\infty }\in \ell _{p^*}^{mid}( Y^*) .$
\item There exist a constant $C>0$ such that 
\begin{equation}\label{equ501}
\Vert ( \varphi_T(x_j,y_j^*) ) _{j=1}^{k }\Vert _{1}\leq C\Vert
( x_{j}) _{j=1}^{k }\Vert _{p}\Vert (y_{j}^{\ast }) _{j=1}^{k }\Vert _{mid,p^*}.
\end{equation}
for any $( x_{j}) _{j=1}^{k}\subset X$ and $( y_{j}^{\ast })_{j=1}^{k}\subset Y^{\ast }.$
\end{enumerate}
Moreover, $d_{p}^{mid}(T)=\Vert \widehat{\varphi_T } \Vert=\inf \{ C: \ (\ref{equ500}) \ holds \}=\inf \{ C: \ (\ref{equ501}) \ holds \} $.
\end{proposition}
%\begin{remark} Let $T\in \mathcal{L}(X;Y)$. Then $T\in \mathcal{D}^{mid}(X;Y)$ if and only if $\varphi_T$ satisfy Proposition.\end{remark}
\begin{theorem}$(\mathcal{D}_{p}^{mid}(X;Y),d_{p}^{mid}(\cdot ))$ is a Banach operator ideal.
\end{theorem}
\begin{proof}
Using the abstract framework, notation and language \cite{transformation}, we find that a linear operator $T$ is  strongly $p$-summing if and only if $\varphi_T$ is $(\ell_{p}(.)\ell_{p^*}^{mid}(.);\ell_1)$-summing. Since $1/p+1/p^*=1$  we obtain $$\Vert (\lambda_j^1.\lambda_j^2)_{j=1}^{\infty}\Vert_1\leq \Vert (\lambda_j^1)_{j=1}^{\infty}\Vert_{\ell_p(\mathbb{K})} \Vert (\lambda_j^2)_{j=1}^{\infty}\Vert_{\ell_{p^*}(\mathbb{K})}=\Vert (\lambda_j^1)_{j=1}^{\infty}\Vert_{\ell_p(\mathbb{K})} \Vert (\lambda_j^2)_{j=1}^{\infty}\Vert_{\ell_{p^*}^{mid}(\mathbb{K})}.$$Therefor,
 $ \  \ell_{p}(\mathbb{K}) \ell_{p^*}^{mid}(\mathbb{K})= \ell_{p}(\mathbb{K}) \ell_{p^*}(\mathbb{K}) \overset{1}{\hookrightarrow }\ell_{1}.$ In addition, all the sequence classes involved are linearly stable (see \cite[Proposition 1.10]{mid1}). So, form \cite[Theorem 3.6]{transformation} it follows that  $( \mathcal{D}^{mid}_{p},d^{mid}_{ p}(.))$ is a Banach operators ideal.
\end{proof}
In the next theorem, we will show that in fact the absolutely mid $p$-summing linear operators is the adjoint of  strongly mid $p$-summing linear operators that will be useful throughout our section.
\begin{theorem}\label{theequ} Let $1<p\leq \infty$, $T\in \mathcal{L}(X;Y)$. Then  $T\in \mathcal{D}_p^{mid}(X;Y)$ if and only if $T^*\in \Pi_{p^*}^{mid}(Y^*;X^*)$. Moreover, $d_p^{mid}(T)=\pi_{p^*}^{mid}(T^*)$.
\end{theorem}
\begin{proof}Assume that $T\in \mathcal{D}_p^{mid}(X;Y)$. For $(x_{j})_{j=1}^{\infty} \in  \ell _{p}( X)$ and $(y_{j}^{\ast })_{j=1}^{\infty} \in \ell _{p^*}^{mid}( Y^*)$. We have 
\begin{equation*}
\Vert ( \langle T( x_{j}),y_{j}^*\rangle ) _{j=1}^{\infty}\Vert _{1}=\Vert ( \langle T^{\ast }( y_{j}^{\ast }),x_{j}\rangle ) _{j=1}^{\infty}\Vert _{1}\leq d_p^{mid}(T) \Vert( x_{j}) _{j=1}^{\infty}\Vert _{p}\Vert ( y_{j}^{\ast}) _{j=1}^{\infty}\Vert _{mid,p^*}
\end{equation*}
by taking the supremum over the unit ball in $\ell _{p}( X) $ we obtain 
$\Vert (T^{\ast }( y_{j}^{\ast }) )_{j=1}^{\infty}\Vert _{p^{\ast }}\leq C\Vert (y_{j}^{\ast}) _{j=1}^{\infty}\Vert _{mid,p^*}
$. Therefore, $T^{\ast }\colon Y^{\ast }\rightarrow X^{\ast }$ is absolutely mid $p^*$-summing and $d_p^{mid}(T)\geq \pi_p^{mid}(T^*)$. Conversely, let $T^*\in \Pi_{p^*}^{mid}(Y^*;X^*)$. For $(x_{j})_{j=1}^{\infty} \in  \ell _{p}( X)$ and $(y_{j}^{\ast })_{j=1}^{\infty} \in \ell _{p^*}^{mid}( Y^*)$. By H\"{o}lder's inequality we have 
$$
\sum_{j=1}^{\infty}| \langle T(x_{j}),y_{j}^{\ast}\rangle | \leq \Vert (x_{j})_{j=1}^{\infty}\Vert_{p}\Vert (T^*(y_{j}^*))_{j=1}^{\infty}\Vert _{p^{\ast }} \leq \pi_p^{mid}(T^*) \Vert (x_{j})_{j}^\infty\Vert _{p}\Vert ( y_{j}^{\ast})) _{j=1}^{\infty}\Vert_{mid,p^{\ast }}.
$$Therefore $T$ is strongly mid $p$-summing and $d_p^{mid}(T)=\pi_p^{mid}(T^*)$.
\end{proof}
As a consequence, we obtain the following corollary which is a straightforward consequence of the preceding theorem and Theorem 2.8 in \cite{mid1}.
\begin{corollary}
Let $1\leq p<\infty$, $T\in \mathcal{L}(X;Y)$. If $T^*\in \mathcal{D}_{p^*}^{mid}(Y^*;X^*)$ then $T\in \Pi_p^{mid}(X;Y)$.
\end{corollary}
Recall that an operator ideal $\mathcal{I}(X;Y)$ is surjective if $T\in\mathcal{I}(X;Y) $ whenever $T\circ Q \in\mathcal{I}(X_0;Y)$, where $Q:X_0\rightarrow X$ is the quotient map.
As consequence, we obtain the following proposition which is a straightforward consequence of Theorem \ref{theequ}, and \cite[Proposition 2.8]{mid1} and \cite[Section 4]{Pie80}.
\begin{proposition}The operator ideal $(\mathcal{D}_p^{mid},d^{mid}_p)$ is surjective.
\end{proposition}
 The Dvoretzky–Rogers theorem (see \cite{DJT}) states that the Banach space $X$ is ﬁnite dimensional if, and only if, $id_X$ is $p$-summing. Now we will give a results in this context.
 
We say that a Banach space $X$ is a weak mid $p$-space (or $p$-Dunford-Pettis property see \cite{KS}) if $\ell^{mid}_p(X)=\ell_p^w(X)$  and it is a strong mid $p$-space if $\ell^{mid}_p(X)=\ell_p(X)$.

The next result is a reformulation of \cite[Theorem 2.7]{mid1}.
\begin{theorem}\label{theorema1} The following are equivalent:\\
1) $X$ is a strong mid $p$-space.\\
2) $id_{X}\in \Pi_{p}^{mid}(X;X)$. \\
3) $\mathcal{L}(X;Y)=\Pi_p^{mid}(X;Y)$ for every Banach space $Y$.\\
4) $\mathcal{L}(Y;X)=\Pi_p^{mid}(Y;X)$ for every Banach space $Y$.\\
5) $X$ is a subspace of $L_p(\mu)$ for some Borel measure $\mu$.
\end{theorem}
\begin{corollary} If $X^{**}$ is a strong mid $p$-space then $X$ is a strong mid $p$-space.
\end{corollary}
\begin{proof}
Is a direct consequence of Theorem \ref{theorema1} and \cite[Proposition 2.8]{mid1}.
\end{proof}
\begin{theorem}\label{theorema2} The following are equivalent:\\
1) $X^*$ is a strong mid $p^*$-space.\\
2) $id_{X}\in \mathcal{D}_{p}^{mid}(X;X)$.\\
3)  $\mathcal{L}(X;Y)=\mathcal{D}_p^{mid}(X;Y)$ for every Banach space $Y$.\\ 
4)  $\mathcal{L}(Y;X)=\mathcal{D}_p^{mid}(Y;X)$ for every Banach space $Y$.\\
5)  $id_{X^*}\in \Pi_{p^*}^{mid}(X^*;X^*)$.\\
6)  $\mathcal{L}(X^*;Y)=\Pi_{p^*}^{mid}(X^*;Y)$ for every Banach space $Y$.\\ 
7)  $\mathcal{L}(Y;X^*)=\Pi_{p^*}^{mid}(Y;X^*)$ for every Banach space $Y$.\\ 
8)  $X^*$ is a subspace of $L_{p^*}(\mu)$ for some Borel measure $\mu$.
\end{theorem}
\begin{proof} By Theorem \ref{theequ} and Theorem \ref{theorema1}, we have
$$\begin{tabular}{ c c c c c c c}
  &   &   &   & 7 &   &   \\ 
  &   &   &   & $\Updownarrow$ &   &   \\ 
8 & $\Leftrightarrow$ & 1 & $\Leftrightarrow$ & 5 & $\Leftrightarrow$ & 6 \\ 
  &   &   &   & $\Updownarrow$ &   &   \\ 
  &   & 3 & $\Leftrightarrow$ & 2 & $\Leftrightarrow$ & 4 \\ 
\end{tabular} $$

\end{proof}
\begin{theorem}\label{2.10}\cite[Theorem 2.6]{mid1}\cite[Theorem 3.7]{KS} The following are equivalent:\\
1) $X$ has the $p$-Dunford-Pettis property.\\
2) $\Pi_{p}^{mid}(X;Y)=\Pi_{p}(X;Y)$ for every Banach space $Y$.\\ 
3) $\Pi_{p}^{mid}(X;\ell_{p})=\Pi_{p}(X;\ell_{p})=\mathcal{L}(X;\ell_{p})$.\\
4) $\mathcal{D}_{p^*}(\ell_{p^*};X^*)=\mathcal{L}(\ell_{p^*};X^*)$.
\end{theorem}
\begin{theorem}\label{theorema4}The following are equivalent:\\
1) $X^*$ has the $p$-Dunford-Pettis property.\\
2) $\mathcal{D}_{p^*}^{mid}(Y;X)=\mathcal{D}_{p^*}(Y;X)$ for every Banach space $Y$.\\
3) $\Pi_{p}^{mid}(X^*;Y^*)=\Pi_{p}(X^*;Y^*)$ for every Banach space $Y$.\\
4) $\Pi_{p}^{mid}(X^*;\ell_{p})=\Pi_{p}(X^*;\ell_{p})=\mathcal{L}(X^*;\ell_{p})$.\\ 
5) $\mathcal{D}_{p^*}^{mid}(\ell_{p^*};X)=\mathcal{D}_{p^*}(\ell_{p^*};X)=\mathcal{L}(\ell_{p^*};X)$.\\
6) $\mathcal{D}_{p^*}(\ell_{p^*};X)=\mathcal{L}(\ell_{p^*};X)$.
\end{theorem}
\begin{proof}
follows from Theorem \ref{theequ} and Theorem \ref{2.10}.
\end{proof}
\begin{corollary} $X^{**}$ has the $p$-Dunford-Pettis property if and only if $X$ has the $p$-Dunford-Pettis property.
\end{corollary}
\begin{corollary}\cite[Corollary 3.9]{KS} If $X^{*}$ has the $p$-Dunford-Pettis property, so does $X$.
\end{corollary}
\begin{theorem}$X^{*}$ has the $p$-Dunford-Pettis property if and only if  $X$ has the $p$-Dunford-Pettis property.
\end{theorem}


%\begin{thebibliography}
%%include bibliography file or list
%\end{thebibliography}
\end{document}
